# Seasonal Changes in Poleward Atmospheric Heat Transport Under Increased CO2

Lily Caroline Hahn<sup>1</sup>, Kyle Armour<sup>1</sup>, David S. Battisti<sup>1</sup>, Aaron Donohoe<sup>1</sup>, and Robert Fajber<sup>1</sup>

<sup>1</sup>University of Washington

July 8, 2023

#### Abstract

Arctic warming under increased CO2 peaks in winter, but is influenced by summer forcing via seasonal ocean heat storage. Yet changes in atmospheric heat transport into the Arctic have mainly been investigated in the annual mean or winter, with limited focus on other seasons. We investigate the full seasonal cycle of poleward heat transport modelled with increased CO2 or with individually applied Arctic sea-ice loss and global sea-surface warming. We find that a winter reduction in dry heat transport is driven by Arctic sea-ice loss and warming, while a summer increase in moist heat transport is driven by sub-Arctic warming and moistening. Intermodel spread in Arctic warming controls spread in seasonal poleward heat transport. These seasonal changes and their intermodel spread are well-captured by down-gradient diffusive heat transport. While changes in moist and dry heat transport compensate in the annual-mean, their opposite seasonality may support non-compensating effects on Arctic warming.

#### Hosted file

966945\_0\_art\_1\_rwvtjr.docx available at https://authorea.com/users/556476/articles/652158-seasonal-changes-in-poleward-atmospheric-heat-transport-under-increased-co2

#### Hosted file

966945\_0\_supp\_11121098\_rwqdzk.docx available at https://authorea.com/users/556476/articles/ 652158-seasonal-changes-in-poleward-atmospheric-heat-transport-under-increased-co2

1 2 3 4 5	
6	Seasonal Changes in Poleward Atmospheric Heat Transport Under Increased CO <sub>2</sub>
7	L. C. Hahn <sup>1</sup> , K. C. Armour <sup>1, 2</sup> , D. S. Battisti <sup>1</sup> , A. Donohoe <sup>3</sup> , and R. Fajber <sup>1</sup>
8	<sup>1</sup> Department of Atmospheric Sciences, University of Washington, Seattle, WA, USA
9	<sup>2</sup> School of Oceanography, University of Washington, Seattle, WA, USA.
10 11	<sup>3</sup> Polar Science Center, Applied Physics Laboratory, University of Washington, Seattle, WA, USA.
12	
13 14	Corresponding author: Lily Hahn ( <u>lchahn@uw.edu</u> )
15	Key Points:
16 17	• Sea-ice loss reduces dry heat transport to the Arctic in winter; sub-Arctic warming increases latent heat transport to the Arctic in summer.
18 19	• Intermodel spread in Arctic warming controls intermodel spread in seasonal heat transport changes.
20 21	• The seasonal pattern of poleward heat transport change is well-captured by down- gradient diffusion of temperature and moisture anomalies.
22 23	

# 24 Abstract

Arctic warming under increased  $CO_2$  peaks in winter, but is influenced by summer forcing via seasonal ocean heat storage. Yet changes in atmospheric heat transport into the Arctic have mainly been investigated in the annual mean or winter, with limited focus on other seasons. We investigate the full seasonal cycle of poleward heat transport modelled with increased  $CO_2$  or with individually applied Arctic sea-ice loss and global sea-surface warming. We find that a

30 winter reduction in dry heat transport is driven by Arctic sea-ice loss and warming, while a

31 summer increase in moist heat transport is driven by sub-Arctic warming and moistening.

32 Intermodel spread in Arctic warming controls spread in seasonal poleward heat transport. These

33 seasonal changes and their intermodel spread are well-captured by down-gradient diffusive heat

transport. While changes in moist and dry heat transport compensate in the annual-mean, their

35 opposite seasonality may support non-compensating effects on Arctic warming.

36

# 37 Plain Language Summary

38 The Arctic is warming much faster than the rest of the planet in response to rising greenhouse gas concentrations. Because Arctic warming peaks in winter, many studies have focused on the 39 40 wintertime processes amplifying Arctic warming. However, others have found that summer 41 atmospheric heating also contributes to winter warming by melting sea ice and storing heat in the 42 ocean until it is released to the atmosphere in winter. Here we study changes in all seasons for one source of atmospheric heating in the Arctic-atmospheric heat transport from lower 43 44 latitudes. Using climate model simulations, we find that heat and moisture are transported away from the regions that warm and moisten the most in response to rising greenhouse gas 45 concentrations. The Arctic warms more than lower latitudes in winter, which reduces heat 46 transport to the Arctic in winter. Atmospheric moisture increases most in late summer at lower 47 latitudes, driving increased moisture transport in late summer from lower latitudes to the Arctic. 48 We suggest that changes in heat and moisture transport may impact Arctic warming differently 49 50 due to their opposite seasonality: by producing a larger change in surface solar reflectivity, summer changes in moisture transport may outweigh winter changes in heat transport. 51

52

# 53 **1 Introduction**

The Arctic has warmed as much as four times faster than the global mean in recent decades (Chylek et al., 2022; Hahn et al., 2021; Rantanen et al., 2022), motivating research to understand what produces this Arctic-amplified warming pattern. Local climate forcing and feedbacks associated with sea-ice loss are thought to contribute most to Arctic-amplified warming (Hwang et al., 2011; Kay et al., 2012; Stuecker et al, 2018). In contrast, annual-mean atmospheric heat transport (AHT) from lower latitudes to the Arctic changes little under CO<sub>2</sub> forcing in comprehensive climate models, suggesting that it makes a small contribution to Arctic

warming (Goosse et al., 2018; Pithan and Mauritsen, 2014). However, this small change in total 61 poleward AHT reflects compensation between larger changes in decreased dry heat transport and 62 increased latent heat transport, which itself has been highlighted as a major contributor to Arctic 63 warming (e.g., Alexeev et al., 2005; Armour et al., 2019; Feldl and Merlis, 2021; Graversen and 64 Wang, 2009; Merlis and Henry, 2018; Woods and Caballero, 2016). By separating each of these 65 components in the latest generation of climate models, Hahn et al. (2021) find that increased 66 latent heat transport is the third largest contributor to Arctic-amplified warming, after local 67 albedo and lapse-rate feedbacks. Others suggest that increased latent heat transport will outweigh 68 69 decreased dry heat transport by contributing a larger greenhouse effect, yielding a net warming effect of projected heat transport changes into the Arctic (Graversen and Burtu, 2016; Graversen 70 and Langen, 2019). 71

Changes in poleward AHT under increased CO<sub>2</sub> forcing have been investigated in the 72 annual-mean from a diffusive perspective, in which AHT is proportional to meridional gradients 73 in temperature and moisture (e.g., Armour et al., 2019; Roe et al., 2015). In this perspective, 74 increased latent heat transport to the Arctic with increased CO<sub>2</sub> results from greater moistening at 75 warmer, lower latitudes than at the poles, following the Clausius-Clapeyron relation. This 76 77 amplifies the meridional gradient in moisture and therefore the poleward latent heat transport (Armour et al., 2019; Held and Soden, 2006; Siler et al., 2018). In contrast, stronger warming at 78 the poles than at lower latitudes weakens the meridional temperature gradient and reduces dry 79 80 AHT to the Arctic (Armour et al., 2019; Feldl et al., 2017; Henry et al., 2021). Consistent with this perspective, dry heat transport decreases most in models with larger Arctic feedbacks and 81 warming, suggesting that poleward dry AHT weakens in response to Arctic warming (Hahn et 82 83 al., 2021; Hwang et al., 2011; Pithan and Mauritsen, 2014).

### manuscript submitted to Geophysical Research Letters

84	While heat transport changes have been studied in the annual mean or in specific seasons,
85	the full seasonal cycle of heat transport and its drivers have received less attention. Some studies
86	highlight the role of latent heat transport in winter as a key contributor to winter-amplified Arctic
87	warming (Doyle et al., 2011; Gong et al., 2017; Luo et al., 2017; Woods et al., 2013; Woods and
88	Caballero, 2016), while others emphasize the importance of springtime latent heat transport for
89	preconditioning extreme summer sea-ice melt (Kapsch et al., 2013; Mortin et al., 2016). Under
90	rising CO <sub>2</sub> , climate models project that latent heat transport to the Arctic will increase most in
91	summer, while dry heat transport will decrease most in winter (Kaufman and Feldl, 2022;
92	McCrystall et al., 2021). However, these changes have not been studied across the full range of
93	climate models, and the causes and impacts of this seasonality have not been fully explored.
94	Based on previous studies, we expect that the winter peak in Arctic warming will be
95	directly damped by decreased dry AHT in winter, but indirectly amplified by increased latent
96	AHT in summer, which will enhance the summer ice-albedo feedback, seasonal ocean heat
97	storage, and its winter release to the atmosphere (Chung et al., 2020; Dai et al., 2019; Deser et
98	al., 2010; Manabe and Stouffer, 1980; Screen and Simmonds, 2010b). Moreover, others have
99	found that summer radiative forcing causes a larger annual-mean Arctic warming than the same
100	amount of winter radiative forcing (Bintanja and Krikken, 2016). This suggests that despite its
101	winter amplification, Arctic warming is impacted by year-round changes in poleward AHT.
102	Given the disproportionate impact of atmospheric forcing in different seasons, understanding the
103	seasonality of poleward heat transport will be essential for understanding observed Arctic
104	changes and accurately predicting future Arctic warming.
105	In this study, we explore the seasonal cycle of latent and dry heat transport in climate

106 model simulations with abrupt CO<sub>2</sub> quadrupling and with individually applied sea-surface

warming and Arctic sea-ice loss from a 2°C global warming scenario. These experiments allow
us to explore how the Arctic and lower latitudes contribute to seasonal heat transport changes
and their intermodel spread. To understand these changes, we investigate the utility of a diffusive
perspective for predicting the seasonal evolution of poleward heat transport. We conclude by
considering how seasonality in latent and dry heat transports may mediate their impacts on
Arctic warming.

113 **2 Results** 

#### 114 2.1 Seasonal Changes in Poleward Heat Transport in CMIP6 Models

We analyze seasonality in heat transport using output from 41 fully-coupled climate

116 models participating in the Coupled Model Intercomparison Project phase 6 (CMIP6;

117 Supplementary Table S1; Eyring et al., 2016). We calculate anomalies using abrupt CO<sub>2</sub>

118 quadrupling simulations (*abrupt4xCO2*) in comparison with pre-industrial control (*piControl*)

simulations. As in previous studies (e.g., Hahn et al., 2021), we apply a 21-year running mean to

120 the *piControl* simulations to account for model drift before calculating anomalies between

121 corresponding periods in the *abrupt4xCO2* and *piControl* simulations. We take 31-year averages

122 centered on year-100 after  $CO_2$  quadrupling to compute monthly anomalies.

We calculate the seasonal cycle of AHT convergence using the difference between the net top-of-atmosphere radiation (TOA) and net surface heat flux (SHF), accounting for atmospheric energy and moisture storage terms following Donohoe et al. (2020). We calculate the latent component of the total heat transport ( $AHT_{latent}$ ) using the difference between evaporation (E) and precipitation (P) multiplied by the latent heat of vaporization (L), and calculate the dry component ( $AHT_{dry}$ ) as the residual between the total and latent heat transports:

129 
$$\operatorname{AHT}(\theta) = -2\pi a^2 \int_{\theta}^{90} \cos(\theta) \left[ TOA(\theta) - SHF(\theta) - \frac{1}{g} \int_{0}^{p_s} \frac{d}{dt} \left( c_p T(\theta) + Lq(\theta) \right) dp \right] d\theta ; \quad (1a)$$

130 
$$\operatorname{AHT}_{\operatorname{latent}}(\theta) = -2\pi a^2 \int_{\theta}^{90} L\cos(\theta) \left[ E(\theta) - P(\theta) - \frac{1}{g} \int_{0}^{p_s} \frac{d}{dt} q(\theta) dp \right] d\theta ; \quad (1b)$$

131 
$$AHT_{dry}(\theta) = AHT(\theta) - AHT_{latent}(\theta)$$
, (1c)

where *a* is the radius of Earth,  $\theta$  is latitude, *g* is the acceleration due to gravity,  $c_p$  is the specific heat of air,  $p_s$  is the surface pressure, *T* is the atmospheric temperature, *q* is the specific humidity, and  $\frac{d}{dt}$  is calculated using centered finite differences of monthly-mean data. The atmospheric energy storage does not include a geopotential term because this changes only by thermal expansion, which is accounted for by using the specific heat at constant pressure,  $c_p$  (Donohoe et al., 2020; Trenberth and Stepaniak, 2004). To assess heat transport changes into the Arctic, we focus on AHT at 70°N, and define the Arctic as 70-90°N.

139 In CMIP6 models under abrupt CO<sub>2</sub> quadrupling, changes in latent and dry heat transports exhibit opposite sign and seasonality (Figures 1a, b). Latent heat transport into the 140 141 Arctic increases year-round, with a maximum increase in summer and fall; dry heat transport into the Arctic decreases year-round, with a maximum decrease in winter. As a result, the total 142 143 poleward heat transport increases in summer and decreases in winter. These results are consistent 144 with the seasonal pattern of heat transport change found in a single large-ensemble model 145 (Kaufman and Feldl, 2022) and with a summer maximum in vertically-integrated moisture flux at 70°N found across CMIP6 models (McCrystall et al., 2021). In the annual- and ensemble-146 mean at 70°N, the reduction in dry heat transport (-0.17 PW) overcompensates the increase in 147 148 latent heat transport (0.09 PW) to produce a net negative change (-0.08 PW). Large intermodel spread in the total heat transport change is dominated by intermodel spread in dry heat transport 149 150



153 Figure 1. (a) Seasonal and (b) monthly-mean changes in moist (blue), dry (yellow), and total (grey) atmospheric heat transport (AHT; PW) at 70°N, averaged over 31 years centered on year-100 after CO<sub>2</sub> quadrupling in CMIP6. 154 155 (c,d) Monthly-mean change in AHT calculated from down-gradient diffusion of (c) surface (d) and vertically-156 integrated anomalies in moist and dry static energy in the same simulations. Line plots (b-d) show the ensemblemean change in AHT, and box plots (a) show the minimum, 25<sup>th</sup> percentile, median, 75<sup>th</sup> percentile, and maximum 157 158 change in AHT across CMIP6 models.

159

152

as a result of larger climatological values; when normalized by the climatology, intermodel 160

spread is larger for the relative change in moist heat transport (Figure S1a). Large intermodel 161

spread persists when heat transport changes are normalized by global-mean near-surface 162

warming in each model (Figure S1b). 163

#### 2.2 A Diffusive Perspective on Seasonal Changes in Poleward Heat Transport 164

We next explore what causes this seasonal pattern of heat transport change from a 165

diffusive perspective. Down-gradient diffusion of near-surface temperature and specific humidity 166

has previously been used to explain heat transport changes in the annual mean (e.g., Armour et 167

al., 2019; Flannery, 1984; Frierson et al., 2007; Hwang et al., 2011; Roe et al., 2015). Can 168

diffusive transport also explain the seasonality of heat transport changes? 169

In a diffusive perspective, atmospheric heat transport is assumed to be proportional to the 170

meridional gradient of moist static energy (MSE); this gradient can be calculated separately for 171

the latent energy (Lq) and dry static energy ( $c_pT + gZ$ ; Z is geopotential height) components of 172

- 173 MSE to partition latent and dry heat transports (e.g., Armour et al., 2019; Bonan et al., 2023;
- 174 Siler et al., 2018). Down-gradient diffusion is typically applied to the near-surface MSE,
- 175 eliminating the geopotential term and resulting in:

176 
$$\operatorname{AHT}_{dry}(x) = -2\pi a^2 D(1-x^2) c_p \frac{dT_s}{dx}$$
, (2a)  $\operatorname{AHT}_{latent}(x) = -2\pi a^2 D(1-x^2) L \frac{dq_s}{dx}$ , (2b)

where x is the sine of latitude, D is a diffusivity constant,  $T_s$  is the near-surface temperature, and  $q_s$  is the near-surface specific humidity. This diffusive perpective can also be applied to MSE integrated throughout the Arctic troposphere:

180 
$$\operatorname{AHT}_{dry}(x) = -2\pi a^2 D (1 - x^2) \frac{d}{dx} \frac{1}{p_{trop}} \int_{300 \ hPa}^{p_s} c_p T + g Z \ dp,$$
(3a)

181 
$$AHT_{latent}(x) = -2\pi a^2 D(1-x^2) \frac{d}{dx} \frac{1}{p_{trop}} \int_{300 \ hPa}^{p_s} Lq \ dp,$$
(3b)

182 where we take 300 hPa to be representative of the Arctic tropopause, and  $p_{trop} = \int_{300 hPa}^{p_s} dp$  is 183 the pressure thickness of the troposphere.

We use an annual-mean diffusivity D calculated from the ensemble mean of piControl 184 simulations by setting the total diffusive heat transport equal to the total heat transport from Eq. 185 (1). This yields a diffusivity of 0.3 W m<sup>-2</sup> K<sup>-1</sup> at 70°N for surface diffusion, similar to values of D 186 diagnosed in previous studies (e.g., Hwang and Frierson, 2010), and a diffusivity of 0.6 W m<sup>-2</sup> K<sup>-</sup> 187 <sup>1</sup> at 70°N for vertically-integrated diffusion. We find the same values of diffusivity when using 188 MSE gradients and total heat transport averaged more broadly from 65 to 75°N. We use these 189 190 values of diffusivity for the CO<sub>2</sub> forcing simulations as well, with the assumption that diffusivity changes are relatively small in these simulations (Armour et al., 2019; Roe et al., 2015). 191 When applied to near-surface anomalies in temperature and humidity from CMIP6 192 models (Eq. 2), down-gradient diffusion does not fully capture the seasonal pattern of heat 193

194 transport change. In line with actual changes (Figure 1b), a diffusive perspective predicts

195 increased moisture transport in summer and decreased dry heat transport in winter (Figure 1c). However, near-surface diffusion also predicts decreased moisture transport in winter, and 196 overestimates dry heat transport changes by an order of magnitude. This large decrease in 197 diffusive dry heat transport results from Arctic-amplified near-surface warming that peaks in 198 winter (Figure S2d). Meanwhile, increased poleward moisture transport in summer results from 199 200 warmer preindustrial temperatures at lower latitudes, which moisten more than the Arctic under CO<sub>2</sub> forcing following the Clausius-Clapeyron relation (Figures S2a, e-f). In winter, this initial-201 temperature effect is overcome by Arctic-amplified winter warming, which produces a larger 202 203 moistening at higher latitudes (Figures S2d-f) and decreases the diffusive moisture transport to the Arctic. 204

With near-surface diffusion failing to capture the full seasonal pattern of poleward heat 205 transport changes, we consider diffusion of anomalies in moist static energy integrated 206 throughout the troposphere (Eq. 3). This vertically-integrated diffusion is motivated by the 207 expectation that transient eddies respond to meridional temperature gradients throughout the 208 troposphere, not just at the surface. We expect vertically-integrated diffusion to better predict 209 heat transport particularly in the Arctic, where the full tropospheric temperature response is 210 211 decoupled from the surface response as a result of stable surface inversions in winter (Cronin and Jansen, 2015; Payne et al., 2015). We note that the value of diffusivity we use for the vertically-212 averaged MSE diffusion is approximately twice the magnitude of that used for near-surface 213 214 diffusion as the meridional MSE gradients are generally weaker higher in the atmosphere.

Arctic stability supports peak warming and moistening near the surface in winter under CO<sub>2</sub> forcing, with weaker changes aloft (Figure S3). As a result, vertically-integrated warming is less Arctic-amplified than near-surface warming. This yields a weaker reduction in diffusive dry

218 heat transport that more closely resembles the actual heat transport change (Figure 1d). Also in line with actual heat transport changes, larger vertically-integrated moistening at lower latitudes 219 than in the Arctic produces an almost year-round increase in diffusive latent heat transport that 220 peaks during summer (Figure 1d). This yields smaller differences between the actual heat 221 transport diffusive transport of vertically-integrated anomalies than for surface diffusion (Table 222 S2). We find similar results when diffusive transport is applied exclusively to the lower 223 troposphere (integrated from 1000 to 600 hPa; not shown), where eddy poleward heat flux is 224 projected to change most under CO<sub>2</sub> forcing (Audette et al., 2021; Kaufman and Feldl, 2022). 225 226 In summary, vertically-integrated diffusion captures the magnitude and sign of seasonal heat transport changes better than near-surface diffusion due to a dropoff in Arctic warming with 227 height. The fit between the diffusive and actual heat transports could be improved by allowing 228 for seasonal variations in diffusivity, but even with annually constant diffusivity, seasonality in 229 temperature and humidity anomalies broadly predicts seasonality in heat transport changes. We 230 find that a diffusive perspective is useful for understanding the seasonal pattern of poleward heat 231 transport: initially warmer conditions generate greater moistening at lower latitudes, particularly 232 in summer, and support a summer-amplified increase in poleward latent heat transport, while 233 234 winter-peaking Arctic amplification supports a winter-amplified decrease in poleward dry heat 235 transport.

236

2.3 Relative Roles of Arctic and Lower Latitudes for Seasonal Heat Transport Changes

In Section 2.2, we showed that seasonal changes in moist and dry static energy gradients predict seasonal changes in poleward heat transport, following a diffusive perspective. Lowerlatitude moistening appears to primarily control changes in the moisture gradient and therefore latent heat transport, while Arctic warming appears to primarily control changes in the

#### manuscript submitted to Geophysical Research Letters

temperature gradient and therefore dry heat transport. To better isolate the contributions of the
Arctic and lower latitudes to seasonality in heat transport anomalies, we next analyze
atmosphere-only simulations from the Polar Amplification Model Intercomparison Project
(PAMIP; Smith et al., 2019), which separately prescribe changes in Arctic sea-ice concentration
(SIC) and global sea-surface temperature (SST).

In a control PAMIP simulation, year-2000 SIC and SST are applied. In a sea-ice loss 246 simulation ( $\Delta SIC$ ), Arctic SIC anomalies from a 2°C global warming scenario are applied, while 247 the year-2000 SST is held fixed. In a sea-surface warming simulation ( $\Delta SST$ ), global SSTs from 248 the same 2°C warming scenario are applied while the year-2000 SIC is held fixed. This future 249 warming scenario is a snapshot of the high-emissions RCP8.5 pathway at 2°C of global warming 250 relative to preindustrial control conditions. The *ASIC* simulation also applies SSTs from the 2°C 251 warming scenario in grid points that transition from sea-ice cover to open ocean. We use five 252 PAMIP models with sufficient data to compute seasonal atmospheric heat transport 253 (Supplementary Table S1), and calculate anomalies for the  $\triangle SST$  and  $\triangle SIC$  simulations relative 254 to the control simulation. For each model, we take the average of 100 ensemble members with 255 different initial conditions, and analyze the last 12 months of these 14-month simulations. 256 257 The sum of heat transport changes in the individual PAMIP simulations ( $\Delta SST + \Delta SIC$ ) broadly reproduces the seasonal pattern of heat transport change in the CMIP6 abrupt4xCO2 258 simulations: decreased dry heat transport that peaks in winter, and increased latent heat transport 259 260 that peaks in summer (compare Figures 2a and 2d). The heat transport changes in the PAMIP simulations are smaller than those in the *abrupt4xCO2* simulation, which is expected given the 261

weaker global warming in PAMIP (2°C) compared to *abrupt4xCO2* (5.5°C): a smaller global

263 warming

264



Figure 2. Change in dry (top) and moist (bottom) atmospheric heat transport (AHT; PW) in (a) CO<sub>2</sub> quadrupling simulations in CMIP6 and (b, c) PAMIP simulations that individually apply anomalies in (b) sea-ice concentration ( $\Delta$ SIC) and (c) sea surface temperature ( $\Delta$ SST) from a 2°C global warming scenario. The sum of the  $\Delta$ SIC and  $\Delta$ SST simulations is shown in d). Line plots show  $\Delta$ AHT at 70°N for individual models (grey) and the ensemble mean (black), and contour plots show ensemble-mean  $\Delta$ AHT.

- produces smaller Arctic warming and mid-latitude moistening, weaker changes in meridional 272 temperature and moisture gradients, and therefore weaker changes in poleward heat transport 273 from a diffusive perspective. In addition, the *abrupt4xCO2* simulation shows a larger decrease in 274 dry heat transport in late winter (January-March) than the combined PAMIP simulations. This 275 difference is also predicted by diffusive transport: with greater warming in the Arctic, the 276 seasonal warming maximum shifts from early to late winter (Hahn et al., 2022; Holland and 277 Landrum, 2021; Liang et al., 2022), reducing the meridional temperature gradient and shifting 278 the reduction of dry heat transport into late winter. 279 With the PAMIP simulations reproducing the key seasonal features of heat transport 280
- anomalies found in CMIP6 simulations, we next investigate what controls these features—the
- 282 Arctic or lower latitudes? For dry heat transport, Arctic sea ice changes produce a winter-

283 peaking decrease

(Figures 2a,b) by promoting Arctic-amplified warming in winter (Figures S4b,f). The  $\Delta SST$ 284 simulation produces a more meridionally uniform warming (Figures S4c,g), causing little change 285 in dry heat transport (Figure 2c). In contrast, lower-latitude moistening due to sea-surface 286 warming (Figures S3k,o) largely explains the increase in poleward latent heat transport found in 287 CMIP6 (Figures 2a,c). While sea-surface warming increases poleward latent heat transport in 288 289 late summer and fall (Figure 2c), Arctic sea-ice loss contributes to a lesser degree by reducing latent heat transport in early winter (Figure 2b), producing a summer peak in the net latent heat 290 transport increase (Figure 2d). 291

292 Consistent with previous analysis of how the Arctic and lower latitudes contribute to heat transport in the annual-mean (Audette et al., 2021), we find that Arctic sea-ice loss primarily 293 controls dry heat transport change while lower-latitude moistening primarily controls latent heat 294 transport change. When considering the seasonality of heat transport change, we find that Arctic 295 sea-ice loss also plays a role for latent heat transport by reducing it in early winter to support a 296 summer peak, in combination with lower-latitude moistening. These PAMIP simulations again 297 suggest that heat transport changes are broadly consistent with down-gradient diffusion of moist 298 and dry static energy anomalies, which produces similar results for these simulations (Figure 299 300 S5).

301

# 2.4 Intermodel Spread in Seasonal Heat Transport Changes

We next consider the sources of intermodel spread in seasonal heat transport. A diffusive perspective again provides physical insight: CMIP6 models with larger changes in the meridional gradients of winter temperature and summer moisture tend to show larger changes in dry and latent heat transports, respectively (Figures 3e,g). To investigate what regions control intermodel spread in these gradients and in seasonal heat transport, we examine the meridional structure of

307



**Figure 3.** Anomalies in (a, c) near-surface temperature ( $\Delta T$ ; °C) and (b, d) specific humidity ( $\Delta Q$ ; g/kg) in CO<sub>2</sub> quadrupling simulations in CMIP6 for the top and bottom 25% of models sorted by their change in (a, b) winter dry and (c, d) summer moist atmospheric heat transport (AHT) for individual models (thin lines) and their ensemble means (thick lines). Change in winter dry AHT (PW) versus (e) the gradient in  $\Delta T$  at 70°N (°C/sin $\theta$ ; r<sup>2</sup> = 0.59) and (f) Arctic  $\Delta T$  (°C; r<sup>2</sup> = 0.54), and change in summer moist AHT (PW) versus (g) the gradient in  $\Delta Q$  at 70°N (g/kg/sin $\theta$ ; r<sup>2</sup> = 0.35) and (h) Arctic  $\Delta Q$  (g/kg; r<sup>2</sup> = 0.09) for CO<sub>2</sub> quadrupling simulations in 41 CMIP6 models.

temperature and moisture anomalies for models in the top and bottom quartiles of winter dry 316 (Figures 3a,b) and summer latent (Figures 3c,d) heat transport changes. Intermodel differences in 317 Arctic warming contribute more than lower latitudes to intermodel spread in both latent and dry 318 heat transport change. Models with the largest reduction in winter dry heat transport have more 319 Arctic warming than other models (weakening the meridional temperature gradient; Figure 3a); 320 models with the largest increase in summer latent heat transport have similar midlatitude 321 322 warming but less Arctic warming than other models (steepening the meridional moisture gradient; Figures 3c,d). 323 Negative correlations exist across CMIP6 models between Arctic warming and dry heat 324 325 transport in winter, and to a lesser extent between Arctic moistening and latent heat transport in summer (Figures 3f,h). Correlations are much weaker between heat transport changes and 326 midlatitude temperature and moisture. This indicates that while poleward latent heat transport 327 increases due to lower-latitude moistening, its intermodel spread at 70°N is more strongly 328

329 controlled by the Arctic than lower latitudes. Using near-surface anomalies in temperature and

moisture produces stronger intermodel correlations with heat transport than vertically-integrated
anomalies (Figure S6), likely because intermodel spread in heat transport is dominated by
considerable intermodel spread in Arctic warming, which is surface-trapped. Our results again
support a diffusive understanding of seasonal changes in poleward heat transport, and suggest a
key role for the Arctic in generating intermodel spread, particularly for dry heat transport.

335

## 3 Conclusions

We investigate the seasonal cycle of poleward atmospheric heat transport change in 336 CMIP6 models, the relative roles of Arctic sea-ice loss and sub-Arctic warming in driving those 337 338 changes, and the extent to which heat transport changes can be understood from a diffusive transport perspective. We find a summer maximum in increased latent heat transport and a 339 winter maximum in decreased dry heat transport under CO<sub>2</sub> forcing. While down-gradient 340 diffusion of near-surface anomalies in moist static energy overestimates the extent to which heat 341 transport is reduced in winter, diffusion of vertically-integrated anomalies more accurately 342 predicts the seasonal pattern and magnitude of heat transport changes. PAMIP simulations that 343 isolate the role of Arctic sea-ice loss versus global sea-surface warming also demonstrate that a 344 diffusive perspective can be used to understand seasonality in heat transport. While Arctic sea-345 346 ice loss is responsible for the winter-amplified reduction in dry heat transport, the summeramplified increase in latent heat transport is primarily controlled by sub-Arctic warming and 347 moistening, with a smaller contribution from Arctic sea-ice loss damping latent heat transport in 348 349 early winter. Lastly, we find that Arctic warming differences between models are the dominant contributor to intermodel spread in poleward heat transport in CMIP6 models under CO<sub>2</sub> 350 351 quadrupling, again in line with a diffusive perspective.

352 Our results suggest that a diffusive transport model is an effective way to understand changes in poleward heat transport seasonally, in addition to the annual mean. Diffusive 353 transport offers intuition into how poleward heat transport will evolve over time based on future 354 temperature and moisture gradients: for example, we expect that the migration of the seasonal 355 maximum in Arctic warming from early to late winter (Hahn et al., 2022; Holland and Landrum, 356 357 2021; Liang et al., 2022) will also shift the reduction in poleward heat transport from early to late winter. A diffusive perspective also indicates that heat transport acts to dampen intermodel 358 spread in Arctic warming, as models with weaker Arctic warming exhibit more poleward heat 359 360 transport.

While past literature has focused on the contribution of latent heat transport in winter to 361 the winter peak in Arctic warming (Doyle et al., 2011; Gong et al., 2017; Luo et al., 2017; 362 Woods et al., 2013; Woods and Caballero, 2016), we find that CMIP6 models predict the largest 363 increases in latent heat transport in summer. As these summer changes will be translated into 364 winter warming via seasonal ocean heat storage and sea-ice thinning (Chung et al., 2020; Dai et 365 al., 2019; Deser et al., 2010; Manabe and Stouffer, 1980; Screen and Simmonds, 2010b), future 366 research should investigate the impact of heat transport changes in summer as well as winter. 367 368 Graversen and Langen (2019) posit that Arctic warming from increased latent heat transport will outweigh cooling from decreased dry heat transport by producing a larger greenhouse effect and 369 a larger sea-ice albedo change for a positive versus negative forcing. We hypothesize that 370 371 opposite seasonality in latent and dry heat transports is another reason to expect noncompensating effects of changes in atmospheric heat transport on Arctic warming. Specifically, 372 we expect the summer amplification of a given increase in latent heat transport to cause greater 373 374 Arctic warming than the cooling caused by the same magnitude of decreased dry heat transport

### manuscript submitted to Geophysical Research Letters

375	in winter. This intuition is built on past findings that summer forcing produces more annual-
376	mean warming in the Arctic by supporting a larger albedo feedback than winter forcing (Bintanja
377	and Krikken, 2016). Our results underscore the importance of studying poleward heat transport
378	in all seasons, rather than only the season of peak Arctic warming. Future efforts to understand
379	and predict Arctic climate change should investigate how seasonality in heat transports, as well
380	as other Arctic feedbacks, mediates their effect on Arctic climate change.
381	Acknowledgments
382	We acknowledge high-performance computing support from Cheyenne
383	(doi:10.5065/D6RX99HX) provided by NCAR's Computational and Information Systems
384	Laboratory (2019), sponsored by the National Science Foundation. LCH was supported by the
385	National Science Foundation (NSF) Graduate Research Fellowship Grant DGE-1762114 and the
386	ARCS Foundation Fellowship. LCH and DSB were supported by the Tamaki Foundation. AD,
387	DSB, and KCA were supported by NSF Grant AGS-2019647. KCA was supported by NSF
388	Grant AGS-1752796 and an Alfred P. Sloan Research Fellowship. RF was supported by the
389	NOAA Climate and Global Change Postdoctoral Fellowship programs for the Advancement of
390	Earth System Science (CPAESS) under award NA18NWS4620043B.
391	Data Availability Statement
392	All CMIP and PAMIP data analyzed in this study can be found in the Earth System Grid
393	Federation (ESGF) repository at <u>https://esgf-node.llnl.gov/projects/esgf-llnl/</u> . The CMIP and
394	PAMIP models used in this analysis are listed in Supplementary Table S1.

395 **References** 

- 396 Alexeev, V.A., Langen, P.L., & Bates, J.R. (2005). Polar amplification of surface warming on an
- aquaplanet in "ghost forcing" simulations without sea ice feedbacks. Clim. Dyn. 24, 655–

398 666. <u>https://doi.org/10.1007/s00382-005-0018-3</u>

- 399 Armour, K. C., Siler, N., Donohoe, A., & Roe, G. H. (2019). Meridional Atmospheric Heat
- 400 Transport Constrained by Energetics and Mediated by Large-Scale Diffusion. J.
- 401 *Clim.* 32(12), 3655-3680. <u>https://doi.org/10.1175/JCLI-D-18-0563.1</u>
- 402 Audette, A., Fajber, R. A., Kushner, P. J., Wu, Y., Peings, Y., Magnusdottir, G., et al.
- 403 (2021). Opposite responses of the dry and moist eddy heat transport into the Arctic in the
- 404 PAMIP simulations. *Geophys. Res. Lett.*, 48,
- 405 e2020GL089990. <u>https://doi.org/10.1029/2020GL089990</u>
- Bintanja, R., & Krikken, F. (2016). Magnitude and pattern of Arctic warming governed by the
  seasonality of radiative forcing. *Sci Rep*, 6, 38287. https://doi.org/10.1038/srep38287
- Bonan, D. B., Siler, N., Roe, G. H., & Armour, K. C. (2023). Energetic Constraints on the
- 409 Pattern of Changes to the Hydrological Cycle under Global Warming. J. Climate, 36, 3499–
- 410 3522, <u>https://doi.org/10.1175/JCLI-D-22-0337.1</u>.
- 411 Chung, E.-S., Ha, K.-J., Timmermann, A., Stuecker, M. F., Bodai, T., & Lee, S.-K. (2021). Cold-
- season Arctic amplification driven by Arctic ocean-mediated seasonal energy
- 413 transfer. *Earth's Future*, 9, e2020EF001898. <u>https://doi.org/10.1029/2020EF001898</u>
- 414 Chylek, P., Folland, C., Klett, J. D., Wang, M., Hengartner, N., Lesins, G., & Dubey, M.
- 415 K. (2022). Annual mean Arctic Amplification 1970–2020: Observed and simulated by
- 416 CMIP6 climate models. *Geophysical Research Letters*, 49,
- 417 e2022GL099371. <u>https://doi.org/10.1029/2022GL099371</u>

- 418 Cronin, T. W., & Jansen, M. F. (2015). Analytic radiative-advective equilibrium as a model for
- 419 high-latitude climate. *Geophys. Res. Lett.* 43, 449–457.

420 <u>https://doi.org/10.1002/2015GL067172</u>

- 421 Dai, A., Luo, D., Song, M. et al. Arctic amplification is caused by sea-ice loss under increasing
- 422 CO<sub>2</sub>. Nat Commun **10**, 121 (2019). <u>https://doi.org/10.1038/s41467-018-07954-9</u>
- 423 Deser, C., Tomas, R., Alexander, M., & Lawrence, D. (2010). The Seasonal Atmospheric
- 424 Response to Projected Arctic Sea Ice Loss in the Late Twenty-First Century. J. Clim. 23(2),
- 425 333-351. <u>https://doi.org/10.1175/2009JCLI3053.1</u>
- 426 Donohoe, A., Armour, K. C., Roe, G. H., Battisti, D. S., & Hahn, L. (2020). The partitioning of
- 427 meridional heat transport from the Last Glacial Maximum to CO<sub>2</sub> quadrupling in coupled
- 428 climate models. J. Clim. 33(10), 4141-4165. <u>https://doi.org/10.1175/JCLI-D-19-0797.1</u>
- 429 Doyle, J. G., G. Lesins, C. P. Thackray, C. Perro, G. J. Nott, T. J. Duck, R. Damoah, and J. R.
- 430 Drummond (2011). Water vapor intrusions into the high Arctic during winter. *Geophys. Res.*
- 431 *Lett.*, 38, L12806. doi:<u>10.1029/2011GL047493</u>.
- 432 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., and Taylor, K. E.
- 433 (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6)
- 434 experimental design and organization. *Geosci. Model Dev.*, 9, 1937–1958.
- 435 https://doi.org/10.5194/gmd-9-1937-2016, 2016.
- 436 Feldl, N., S. Bordoni, and T. M. Merlis (2017). Coupled High-Latitude Climate Feedbacks and
- 437 Their Impact on Atmospheric Heat Transport. J. Climate, 30, 189–
- 438 201. <u>https://doi.org/10.1175/JCLI-D-16-0324.1</u>.

- 439 Feldl, N., & Merlis, T. M. (2021). Polar amplification in idealized climates: The role of ice,
- 440 moisture, and seasons. *Geophysical Research Letters*, 48, e2021GL094130.
- 441 https://doi.org/10.1029/2021GL094130
- 442 Flannery, B. P. (1984). Energy balance models incorporating transport of thermal and latent
- 443 energy. J. Atmos. Sci., 41, 414–421. https://doi.org/10.1175/1520-0469(1984)041,0414:
- 444 EBMITO.2.0.CO;2.
- 445 Frierson, D. M. W., Held, I. M., & Zurita-Gotor, P. (2007). A gray-radiation aquaplanet moist
- 446 GCM. Part II: Energy transports in altered climates. J. Atmos. Sci., 64, 1680–1693.
- 447 https://doi.org/10.1175/ JAS3913.1.
- 448 Gong, T., S. Feldstein, & S. Lee (2017). The Role of Downward Infrared Radiation in the Recent
- 449 Arctic Winter Warming Trend. J. Climate, 30, 4937–4949. <u>https://doi.org/10.1175/JCLI-D-</u>
  450 16-0180.1.
- 451 Goosse, H., Kay, J. E., Armour, K. C., Bodas-Salcedo, A., Chepfer, H., Docquier, D., et
- 452 al. (2018). Quantifying climate feedbacks in polar regions. *Nat. Commun.*,
- 453 9, 1919. <u>https://doi.org/10.1038/s41467-018-04173-0</u>
- 454 Graversen, R.G. and Burtu, M. (2016). Arctic amplification enhanced by latent energy transport
- 455 of atmospheric planetary waves. Q.J.R. Meteorol. Soc., 142, 2046-
- 456 2054. <u>https://doi.org/10.1002/qj.2802</u>
- 457 Graversen, R. G., & Langen, P. L. (2019). On the Role of the Atmospheric Energy Transport in 2
- 458 × CO2–Induced Polar Amplification in CESM1. J. Climate, 32, 3941–
- 459 3956. <u>https://doi.org/10.1175/JCLI-D-18-0546.1</u>.
- 460 Graversen, R.G., & Wang, M. Polar amplification in a coupled climate model with locked
- 461 albedo. *Clim Dyn* **33**, 629–643 (2009). https://doi.org/10.1007/s00382-009-0535-6

- 462 Hahn, L. C., Armour, K. C., Zelinka, M. D., Bitz, C. M., & Donohoe, A. (2021). Contributions to
- 463 Polar Amplification in CMIP5 and CMIP6 Models. *Front. in Earth Sci.* 9, 710036.

464 https://doi.org/10.3389/feart.2021.710036

- Hahn, L. C., Armour, K. C., Battisti, D. S., Eisenman, I., & Bitz, C. M. (2022). Seasonality in
- 466 Arctic Warming Driven by Sea Ice Effective Heat Capacity, *Journal of Climate*, 35(5), 1629-
- 467 1642. <u>https://doi.org/10.1175/JCLI-D-21-0626.1</u>
- Held, I. M., & Soden, B. J. (2006). Robust responses of the hydrological cycle to global
- 469 warming. J. Climate, 19, 5686–5699. <u>https://doi.org/10.1175/JCLI3990.1</u>
- 470 Henry, M., T. M. Merlis, N. J. Lutsko, and B. E. J. Rose (2021). Decomposing the Drivers of
- 471 Polar Amplification with a Single-Column Model. J. Climate, 34, 2355–
- 472 2365. <u>https://doi.org/10.1175/JCLI-D-20-0178.1</u>
- 473 Holland, M. M., & Landrum, L. (2021). The Emergence and Transient Nature of Arctic
- 474 Amplification in Coupled Climate Models. *Front. Earth Sci.* 9:719024.
- 475 https://doi.org/10.3389/feart.2021.719024
- 476 Hwang, Y.-T., and Frierson, D. M. W. (2010). Increasing atmospheric poleward energy transport
- 477 with global warming. *Geophys. Res. Lett.*, 37, L24807. doi:<u>10.1029/2010GL045440</u>.
- 478 Hwang, Y.-T., Frierson, D. M. W., & Kay, J. E. (2011). Coupling between Arctic feedbacks and
- 479 changes in poleward energy transport. *Geophysical Research Letters*, 38,
- 480 L17704. <u>https://doi.org/10.1029/2011GL048546</u>
- 481 Kapsch, ML., Graversen, R. & Tjernström, M. (2013). Springtime atmospheric energy transport
- and the control of Arctic summer sea-ice extent. *Nature Clim Change* 3, 744–748.
- 483 <u>https://doi.org/10.1038/nclimate1884</u>

- 484 Kaufman, Z. S., & Feldl, N. (2022). Causes of the Arctic's Lower-Tropospheric Warming
- 485 Structure, Journal of Climate, 35(6), 1983-2002. <u>https://doi.org/10.1175/JCLI-D-21-0298.1</u>
- 486 Kay, J. E., Holland, M. M., Bitz, C. M., Blanchard-Wrigglesworth, E., Gettelman, A., Conley,
- 487 A., & Bailey, D. (2012). The influence of local feedbacks and northward heat transport on
- the equilibrium Arctic climate response to increased greenhouse gas forcing. J.
- 489 *Climate*, *25*(16), 5433–5450. <u>https://doi.org/10.1175/jcli-d-11-00622.1</u>
- 490 Liang, YC., Polvani, L.M. & Mitevski, I. (2022). Arctic amplification, and its seasonal
- 491 migration, over a wide range of abrupt CO<sub>2</sub> forcing. *npj Clim Atmos Sci*, 5, 14.
- 492 https://doi.org/10.1038/s41612-022-00228-8
- 493 Luo, B., Luo, D., Wu, L., Zhong, L., & Simmonds, I. (2017). Atmospheric circulation patterns
- 494 which promote winter arctic sea ice decline. *Environmental Research Letters*, 12(5),
- 495 054017. <u>https://doi.org/10.1088/1748-9326/aa69d0</u>
- 496 Manabe, S., & Stouffer, R. J. (1980). Sensitivity of a global climate model to an increase of
- 497  $CO_2$  concentration in the atmosphere. J. Geophys. Res. 85(C10), 5529-5554.
- 498 <u>https://doi.org/10.1029/JC085iC10p05529</u>
- 499 McCrystall, M.R., Stroeve, J., Serreze, M. et al. New climate models reveal faster and larger
- 500 increases in Arctic precipitation than previously projected. *Nat Commun* 12, 6765 (2021).
- 501 https://doi.org/10.1038/s41467-021-27031-y
- 502 Merlis, T. M., and M. Henry (2018). Simple Estimates of Polar Amplification in Moist Diffusive
- 503 Energy Balance Models. J. Climate, 31, 5811–5824. <u>https://doi.org/10.1175/JCLI-D-17-</u>
- 504 <u>0578.1</u>.

- 505 Mortin, J., Svensson, G., Graversen, R. G., Kapsch, M.-L., Stroeve, J. C., and Boisvert, L.
- 506 N. (2016), Melt onset over Arctic sea ice controlled by atmospheric moisture
- 507 transport, *Geophys. Res. Lett.*, 43, 6636–6642, doi:<u>10.1002/2016GL069330</u>.
- 508 Payne, A. E., Jansen, M. F., & Cronin, T. W. (2015). Conceptual model analysis of the influence
- of temperature feedbacks on polar amplification. *Geophys. Res. Lett.* 42, 9561–9570.
- 510 https://doi.org/10.1002/2015GL065889
- 511 Pithan, F., & Mauritsen, T. (2014). Arctic amplification dominated by temperature feedbacks in
- 512 contemporary climate models. *Nat. Geosci.* 7, 181–184. <u>https://doi.org/10.1038/ngeo2071</u>
- 513 Rantanen, M., Karpechko, A.Y., Lipponen, A. et al. (2022). The Arctic has warmed nearly four
- times faster than the globe since 1979. *Commun Earth Environ*, 3, 168.
- 515 https://doi.org/10.1038/s43247-022-00498-3
- 516 Roe, G., Feldl, N., Armour, K. et al. (2015). The remote impacts of climate feedbacks on
- regional climate predictability. *Nature Geosci*, 8, 135–139. https://doi.org/10.1038/ngeo2346
- 518 Screen, J. A., & Simmonds, I. (2010b). Increasing fall-winter energy loss from the Arctic Ocean
- and its role in Arctic temperature amplification. *Geophys. Res. Lett.* 37, L16707.
- 520 <u>https://doi.org/10.1029/2010GL044136</u>
- 521 Siler, N., Roe, G. H., & Armour, K. C. (2018). Insights into the Zonal-Mean Response of the
- 522 Hydrologic Cycle to Global Warming from a Diffusive Energy Balance Model. *Journal of*
- 523 *Climate*, *31*(18), 7481-7493. <u>https://doi.org/10.1175/JCLI-D-18-0081.1</u>
- 524 Smith, D. M., Screen, J. A., Deser, C., Cohen, J., Fyfe, J. C., García-Serrano, J., et al. (2019).
- 525 The Polar Amplification Model Intercomparison Project (PAMIP) contribution to CMIP6:
- 526 investigating the causes and consequences of polar amplification. *Geosci. Model Dev.* 12,
- 527 1139–1164. https://doi.org/10.5194/gmd-12-1139-2019

529 forcing and feedbacks. *Nat. Clim. Change*, 8, 1076–1081. <u>https://doi.org/10.1038/s41558-</u>

530 <u>018-0339-y</u>

- 531 Trenberth, K.E. and Stepaniak, D.P. (2004), The flow of energy through the earth's climate
- 532 system. Q.J.R. Meteorol. Soc., 130: 2677-2701. <u>https://doi.org/10.1256/qj.04.83</u>
- 533 Woods, C., Caballero, R., & Svensson, G. (2013). Large-scale circulation associated with
- moisture intrusions into the Arctic during winter. *Geophys. Res. Lett.*, 40, 4717–4721.
- 535 doi:<u>10.1002/grl.50912</u>.
- 536 Woods, C., & R. Caballero (2016). The Role of Moist Intrusions in Winter Arctic Warming and
- 537 Sea Ice Decline. J. Climate, 29, 4473–4485. <u>https://doi.org/10.1175/JCLI-D-15-0773.1</u>.

- 539
- 540
- 541
- 542
- 543